Author's response to reviews

Title: Chronic multifocal non-bacterial osteomyelitis in hypophosphatasia mimicking malignancy

Authors:

Hermann J Girschick (Hermann.Girschick@mail.uni-wuerzburg.de)
Etienne Mornet (etienne.mornet@cytogene.uvsq.fr)
Meinrad Beer (beer@roentgen.uni-wuerzburg.de)
Monika Warmuth-Metz (warmuth@neuroradiologie.uni-wuerzburg.de)
Peter Schneider (Schneider_p@klinik.uni-wuerzburg.de)

Version: 4 Date: 21 January 2007

Author's response to reviews:

Dear Lolu da-Silva, assistant editor, BMC series

We would like to send a newly formatted manuscript with the title
Chronic multifocal non-bacterial osteomyelitis in childhood hypophosphatasia mimicking malignancy
to BMC pediatrics for publication.

We have followed the formatting suggestions of the Editorial production team, as suggested.

Sincerely yours
Hermann Girschick, MD
Section of Pediatric Immunology, Rheumatology and Osteology
Universitäts-Kinderklinik
Josef-Schneider-Str.2
97080 Wurzburg
Germany